Author's response to reviews

Title: Parsonage-Turner syndrome following post-exposure prophylaxis

Authors:

Duncan P Fransz (dpfransz@gmail.com)
Casper P Schönhuth (c.schonhuth@vumc.nl)
Tjeerd J Postma (tj.postma@vumc.nl)
Barend J van Royen (bj.vanroyen@vumc.nl)

Version: 2  Date: 6 July 2014

Author's response to reviews: see over
Reviewer's reports:

Reviewer 1; 6 August 2013

Major compulsory revisions: None

Minor essential revisions: There is some overlap in the Background section and Ancillary Investigations section, removing redundancies will make the article more succinct

We agree and have removed the sentences regarding MRI from the Background section.

Discretionary revisions: None

Reviewer 2; 22 April 2014

Major compulsory revisions: None

Minor essential revisions:
Page 2 Line 13 missing space HBV infection
Page 3 Line 7 missing space pain. Although
Page 6 Line 6 missing space indicated against
Page 8 Line 19 missing space Ancillary diagnostic
Page 9 Line 6 missing space out PTS
Page 11 Line 22 missing space and Parsonage
Page 11 Line 24 missing space the occurrence

We have checked the manuscript for missing spaces at the aforementioned locations, but could not find any. Perhaps there was a typesetting mismatch.

Page 11 Line 14 spelling two case reports describe

We have corrected the text as suggested: “Two case reports describe the onset…”

Discretionary revisions:
The authors could publish selected image frames from the MRI study to illustrate the alterations mentioned

We fully agree that MRI images are of illustrative value. However, since we did not perform an MRI scan for the presented case, we feel that the inclusion of MRI images would be a little out of place. The cited study of Gaskin & Helms (2006) presents very illustrative figures, though.